Author's response to reviews

Title: Immunization interventions to interrupt hepatitis B virus mother-to-child transmission: a meta-analysis of randomized controlled trials

Authors:

Hui Jin Prof. (jinhui_hld@163.com)
Zhaoyin Tan Mr. (tan671111@163.com)
Xuefeng Zhang Mr. (njzhangxf@126.com)
Yueyuan Zhao Mrs. (zhaoyy@163.com)
Yaojun Zhao Mrs. (zhaoyj@163.com)
Bei Wang Prof. (wangbeilxb@seu.edu.cn)
Pei Liu Prof. (liupeiseu@163.com)

Version: 3 Date: 20 September 2014

Author's response to reviews:

Dear editors,

Thank you for your suggestions and we have revised the manuscript below:

1. TITLE PAGE:

Please include the email addresses of all authors on the title page. Please ensure that it is the same with the one entered on the submission system.

Answer: Yes, I have added the email addresses of all authors on the title page.

2. ETHICS:

Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Answer: We do not need to consider the problem of ethics, because the article is a meta-analysis without field investigation.

3. ACKNOWLEDGEMENTS:

By way of a section ‘Acknowledgements’, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was
involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible.

The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest wording such as 'We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.'

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

Answer: yes, we have revised the part of acknowledge.

Best Wishes,

Hui Jin